Author's response to reviews

Title: Midwife-led debriefing after operative birth: four to six year follow-up of a randomised trial [ISRCTN24648614]

Authors:

Rhonda Small (r.small@latrobe.edu.au)
Judith Lumley (j.lumley@latrobe.edu.au)
Liesje Toomey (l.toomey@latrobe.edu.au)

Version: 2 Date: 13 November 2005

Author's response to reviews:

As requested, the ISRCTN number for the trial has been obtained and added to the title page of the manuscript and to the title as registered on the submission page.